Author’s response to reviews

Title: What is the job satisfaction and active participation of medical staff in public hospital reform: A study in Hubei province of China

Authors:

Pengqian Fang (pfang@mails.tjmu.edu.cn)  
Zhenni Luo (janelrove@163.com)  
Biyan Wang (wbiyan_1010@163.com)  
Hongxia Gao (banny9566@163.com)  
Wenmin Li (976977873@qq.com)

Version: 7  
Date: 17 September 2014

Author’s response to reviews:

Date: September 17, 2014

Dear Editor and Reviewers,

Firstly, we are really very sorry for not answering your email and not resubmitting the paper in time as requested because there was something wrong with my email box. We would like to thank the editor for giving us a more chance to resubmit the paper, and thank the reviewers for their careful reading and giving us comments and constructive suggestions concerning our manuscript. Here we submit a new version of our manuscript with the revised title “What is the job satisfaction and active participation of medical staff in public hospital reform: A study in Hubei province of China”, which has been modified according to the reviewers’ suggestions. We appreciate for Editors and Reviewers’ warm work earnestly, and hope that the correction will meet with approval.

Once again, thank you very much for your comments and suggestions.

Sincerely yours

Zhenni Luo, Pengqian Fang